Author's response to reviews

Title: Effectiveness of the bucco-lingual technique within a school-based supervised toothbrushing program on preventing caries: a randomized controlled trial

Authors:

Paulo Frazão (pafrazao@usp.br)

Version: 8  Date: 18 February 2011

Author's response to reviews:

My fax number changed; +55-11-30617835

In the second line of second paragraph of Methods the verb 'undergone' was changed to 'undertaken'.